Author's response to reviews

Title: DreamTel; Diabetes Risk Evaluation and Management Tele-monitoring Study

 Authors:

Sheldon Tobe (sheldon.tobe@sunnybrook.ca)
Joan Wentworth (jwentworth@miwayawinhealth.org)
Laurie Ironstand (lironstand@miwayawinhealth.org)
Susan Hartman (shartman@miwayawinhealth.org)
Jackie Hoppe (jhoppe@miwayawinhealth.org)
Judi Whiting (jwhiting@sasktel.net)
Janice Kennedy (jkennedy@miwayawinhealth.org)
Colin McAllister (colin.mcallister@sasktel.sk.ca)
Alex Kiss (alex.kiss@ices.on.ca)
Nancy Perkins (nancy.perkins@sunnybrook.ca)
Lloyd Vincent (lloyd.vincent@sunnybrook.ca)
George Pylypchuk (neph.gbp@sasktel.net)

Version: 2 Date: 13 June 2008

Author's response to reviews:

To Whom it May Concern:

Thank you for considering the DreamTel protocol manuscript for evaluation for BMC Endocrine Disorders. No revisions have been made to this manuscript, but rather, we are re-submitting to BMC as was recommended.

Regards,

Dr. Sheldon Tobe